Author's response to reviews

Title: Recent advances in the diagnosis and treatment of bladder cancer

Authors:

Grace Cheung (dr.grace.cheung@gmail.com)
Arun Sahai (arun.sahai@gstt.nhs.uk)
Michele Billia (michele.billia@gstt.nhs.uk)
Prokar Dasgupta (prokarurol@gmail.com)
Muhammad S Khan (Shamim.khan@gstt.nhs.uk)

Version: 7 Date: 12 November 2012

Author's response to reviews: see over
Monday 12th November 2012

Dear Claire

Please find attached the final version of the bladder cancer manuscript.

As I was having problems with references, there are no tracked changes.

However, you will see that references have been inserted into the introduction and the references should be sequential.

I have also made the other change you requested re TURBT.

Please let me know as soon as possible if there are any changes that need to be made re content.

I am happy to continue to review for general formatting changes.

Grace